

Format for ANSWERING REVIEWERS

October 8, 2014.



Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 13136-review.doc).

Title: EUS ELASTOGRAPHY STRAIN HISTOGRAMS IN THE EVALUATION OF PATIENTS WITH PANCREATIC MASSES

Authors: Opačić D, Rustemović N, Kalauz M, Markoš P, Ostojić Z, Majerović M, Ledinsky I, Višnjić A, Krznarić J, Opačić M.

Name of Journal: *World Journal of Gastroenterology*

ESPS Manuscript NO: 13136

The manuscript has been improved according to the suggestions of reviewers:

- 1 Format has been updated
- 2 Revisions were not asked by the reviewers
- 3 References and typesetting were corrected.

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,

A small, rectangular box containing a handwritten signature in black ink, which appears to be 'M. Opačić'.

A large, stylized handwritten signature in black ink, which appears to be 'M. Opačić', written across the bottom of the page.